Author's response to reviews

Title: A new paradigm in respiratory hygiene: increasing the cohesivity of airway secretions to improve cough interaction and reduce aerosol dispersion

Authors:

- Gustavo Zayas (g.zayas@ualberta.ca)
- John Dimitry (john.dimitry@med.ucalgary.ca)
- Ana Zayas (ana.zayas@ualberta.ca)
- Darryl O'Brien (do1@ualberta.ca)
- Malcolm King (malcolm.king@ualberta.ca)

Version: 6 Date: 1 September 2005

Author's response to reviews:

Added the section "Competing Interests"